Author's response to reviews

Title: Correlates of Opium Use: A Study in Tribal Community of Arunachal Pradesh, India

Authors:

- Himanshu K. Chaturvedi (chaturvedi_icmr@yahoo.com)
- Jagadish Mahanta (jmahanta@hotmail.com)
- Ram C. Bajpai (rampaipai@hotmail.com)
- Arvind Pandey (arvindpandey@icmr.org.in)

Version: 4 Date: 28 February 2013

Author's response to reviews:

Reply of comments

Reviewer 1:

Minor Essential Revisions

The authors sufficiently answered my main methodological and structural questions regarding the study. Nonetheless, there remains a need to make numerous English Language corrections. The authors state that they will work with the editors to make these improvements. The editors may agree to this suggestion, however, the authors may need to work with an English Language specialist who is separate from the journal.

Reply: As advised by the reviewer, the English Language of the manuscript has been corrected.